The Journal of Foot & Ankle Surgery 000 (2023) 1-7



Contents lists available at ScienceDirect

The Journal of Foot & Ankle Surgery



journal homepage: www.jfas.org

Original Research

Exploring the Relationship Between Clinical Presentation in Hallux Valgus and Response to AbobotulinumtoxinA Treatment

Babak Baravarian, DPM¹, Lawrence A. DiDomenico, DPM, FACFAS², Magali Volteau, PhD³, David G. Armstrong, DPM, MD⁴, Robert Silva, PhD⁵

¹ University Foot and Ankle Institute, Los Angeles, CA

² NOMS Ankle and Foot Care Centers, Youngstown, OH

³ Ipsen, Les Ulis, Paris, France

⁴ Keck School of Medicine, University of Southern California, Los Angeles, CA

⁵ Ipsen, Cambridge, MA

ARTICLE INFO

Level of Clinical Evidence: 2

Keywords: angular deviation botulinum toxin foot pain nonsurgical treatment pain severity treatment outcomes

ABSTRACT

The relationship between pain/disability and angular deviation of the hallux valgus (HV), and the impact of orthotic use, laterality, and pain variability on treatment outcomes remain unclear. This was explored in post hoc analyses of a placebo-controlled trial of abobotulinumtoxinA (aboBoNT-A; Dysport[®]) for HV-associated pain (NCT03569098). The primary endpoint was not met in this study (change from baseline Numeric Pain Rating Scale [NPRS] score vs placebo at week 8); however, there was a greater reduction from baseline in mean NPRS score at week 12 with aboBoNT-A 500U versus placebo (p = .06). Adults with painful HV without surgery were randomized (1:1:1) to aboBoNT-A 300U, aboBoNT-A 500U, or placebo. NPRS was completed for 7 days before baseline and weeks 4, 8, and 12. Over-the-counter orthoses were permitted. Participants (N = 186) had a mean [standard deviation, SD] age of 48.2 [13.1] years, 18% (33/186) used orthotics, and 67% (124/186) had bilateral HV. No associations between baseline pain severity and angular deviation were identified. Participants with high pain variability at baseline (SD > 2) had a poorer response to aboBoNT-A treatment than those with less variability. In conclusion, no relationship between HV-related pain/disability and angular deviation was observed.

Plain language summary: A bunion (medical term: hallux valgus) is a common adult foot problem in which the big toe points inward toward the other toes, and this can be painful. How much the big toe points inward (how deformed the foot is) has been linked to the amount of pain the patient experiences. A better understanding of this foot deformity and bunion pain will help doctors and patients to make the right treatment decisions.

A study was completed looking at how injections of a type of botulinum toxin (abobotulinumtoxinA) into specific muscles in the foot may help to reduce bunion pain in patients without surgery. This subsequent analysis of the study data looked at the amount of foot deformity in patients, the bunion pain they experienced, and which factors affected how well abobotulinumtoxinA worked to reduce bunion pain.

The results of this study showed that the amount of foot deformity was not linked to the level of bunion pain. When deciding the best treatment option to relieve bunion pain, it is important that doctors not only consider how deformed the foot is, but also other important factors such as foot pain levels.

© 2023 The Author(s). This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/)

Hallux valgus (HV; bunion) is a progressive foot deformity affecting approximately 23% of adults of 18 to 65 years of age globally. Symptoms

include pain, morphological changes to the foot, and impaired gait and balance (1-3). In HV, the activity of the abductor hallucis muscle of the foot is decreased compared with adductor muscles, and flexor activity of the abductor hallucis increases with severity (4). Hypertonia of the adductor hallucis muscle may result in lateral deviation of the hallux and osseous changes, and development of a pressure-sensitive prominence on the medial side of the first metatarsal that can limit mobility (5,6). Mechanical pressure on this prominence and loss of normal hallux involvement in gait and weight distribution lead to forefoot pain and pain in the lesser toes due to transfer metatarsalgia (7). There is

1067-2516/\$ - see front matter © 2023 The Author(s). This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/) https://doi.org/10.1053/j.jfas.2022.12.016

Financial Disclosure: The study was sponsored by Ipsen, which was involved in the study design, analysis, and interpretation of results.

Conflict of Interest: B.B. and D.G.A. are investigators for the current study and report consultancy (advisory board) for Ipsen. L.A.D. is an investigator for the current study. M.V. is an Ipsen employee. R.S. was an Ipsen employee during the study and manuscript preparation.

Address correspondence to: Robert Silva, PhD, 23A Sycamore St., Millbury, MA 01527. *E-mail address:* Robsilva2188@gmail.com (R. Silva).

substantial evidence that pain avoidance in HV induces compensatory musculoskeletal dysfunction (8,9). Increased severity of HV angle has been correlated with pathomechanical changes in the rear foot and increased pain (10), and knee pain can develop concomitantly in patients with severe HV deformity (11). Compensation also takes place in the contralateral lower limb to accommodate for the presence of HV (8). Given that pathomechanical changes occur secondary to pain avoidance, adequate pain control is a key aspect of HV management.

Physicians, patients, and payors alike need a deeper understanding of how patient and disease characteristics may influence treatment decisions and the response to treatment in HV. Currently, referral for treatment and payor considerations are based on HV angle, pain, and degree of functional impairment (7,12,13). However, there is a general lack of agreement in the HV community regarding the relationship between angular deviation and the degree of foot pain or disability experienced by the patient (14). Indeed, the degree of lateral deviation can overshadow other presenting features, most notably pain severity, but also pain variability, chronicity, and laterality of HV. HV often presents as a bilateral rather than unilateral disease (11), and is often initially managed using orthoses, such as splints, inserts, or braces, to correct foot biomechanics, which are now understood to be largely ineffective (15-17). However, there is little evidence of the impact of these factors on self-reported pain severity and treatment outcomes.

Evidence suggests that HV-associated pain and angular deviation can be reduced following botulinum toxin (BoNT) injections to the affected foot (18-20). BoNTs block the release of presynaptic acetylcholine at the neuromuscular junction to reduce localized muscle tone (21)and block pain signaling, both locally and centrally, via the dorsal root ganglia and spinal cord, to reduce pain sensitization (22,23). A phase 2 trial of abobotulinumtoxinA (aboBoNT-A; Dysport[®], Ipsen, Paris, France; NCT03569098 [ClinicalTrials.gov]) in adults with HV, who had not undergone surgery, reported Numeric Pain Rating Scale (NPRS) scores that were reduced from baseline 12 weeks after intramuscular injection of aboBoNT-A 500 units (U) (24,25). However, this did not reach statistical significance (p = .061) and the primary endpoint (change in least-squares [LS] mean NPRS score from baseline) was not met at 8 weeks. Despite this, the proportion of participants with an at least 20% reduction from their baseline NPRS score was greater with aboBoNT-A 500U than with placebo at week 12 (p = .006). Furthermore, a post hoc analysis revealed that participants in the aboBoNT-A 500U group spent more days with lower NPRS scores than their lowest baseline score (meaningful response), and with NPRS scores ≥ 2 points lower than their mean baseline NPRS score at weeks 8 and 12 compared with placebo (all p < .05) (24,25).

Whereas the aforementioned analysis focused on centrally mediated pain state changes that occur following treatment, the current post hoc analyses of data from this phase 2 trial aimed to determine whether angular deviation in HV affected disability or HV-related pain and what factors affected the response to aboBoNT-A treatment. The hypotheses were that there is a relationship between angular deviation and disability or pain, and that orthotic use, bilateral HV, and pain severity would affect the response to aboBoNT-A treatment.

Materials and Methods

Study Design, Participants, and Treatment

Full study design, eligibility criteria, and treatment administration have been described previously (24). Briefly, the phase 2, parallel-group, multicenter trial had a double-blind phase lasting at least 12 weeks (cycle 1), followed by an open-label phase of up to 24 weeks (cycles 2 and 3) (Fig. 1). Participants had a diagnosis of HV; had not undergone surgery; had no pre-existing medical conditions; had an HV angle of <30° and an intermetatarsal (IM) angle of $<18^{\circ}$ in the great toe; had an NPRS score of \geq 4; and had total scores of >27 on the modified Foot Function Index (mFFI) subscales for pain and disability in the study foot, with pain refractory to shoe modifications, nonsteroidal anti-inflammatory medications, and modification of activities. The use of orthoses on the study foot (except over-the-counter shoe inserts, if used for at least 30 days prior to screening) was not permitted. At the start of cycle 1, participants were randomized 1:1:1 to receive abo-BoNT-A 300U, aboBoNT-A 500U, or placebo divided equally into 4 injections guided by electrical stimulation to the oblique and transverse heads of the adductor hallucis muscle, the flexor hallucis brevis muscle, and the extensor hallucis brevis muscle. After 12 weeks, participants who met retreatment criteria entered the open-label phase and received aboBoNT-A 300U (cycle 2); eligible participants were further retreated with either abo-BoNT-A 300U or aboBoNT-A 500U (investigator's judgment) at least 12 weeks later (cycle 3). Retreatment criteria were participant consent, investigator's clinical judgment, clinically significant foot pain (NPRS score \geq 3) in the preceding 24 hours, and no unacceptable risk to the participant (investigator's judgment). Participants ineligible for retreatment at week 12 were assessed every 4 weeks until retreatment or study end. This study was conducted in accordance with the Declaration of Helsinki, the International Conference on Harmonisation Consolidated Guideline on Good Clinical Practice, and all local regulatory guidelines. Written informed consent was provided by participants prior to study enrollment.

Assessments

Participants, who were blinded to treatment allocation, recorded their pain severity with the NPRS and the mFFI on the 7 consecutive days prior to baseline and at weeks 4, 8, and 12, and the 36-item Short-Form Health Survey (SF-36) at baseline and at weeks 4 and 8 (26-28). At least 4 of the 7 days prior to each time point must have been completed for an assessment to be considered valid for inclusion in analyses. HV angle and IM angle were measured with weightbearing anterior-posterior radiographs, as previously described (24).



Fig. 1. Study design. Retreatment with aboBoNT-A from week 12 was dependent on eligibility determined by the investigator. AboBoNT-A, abobotulinumtoxinA; HV, hallux valgus; IM, intermetatarsal; mFFI, modified Foot Function Index; NPRS, Numeric Pain Rating Scale; SF-36, 36-item Short-Form Health Survey.

B. Baravarian et al. / The Journal of Foot & Ankle Surgery 00 (2023) 1-7

Analyses

All analyses in this manuscript were conducted in a post hoc manner based on a large. placebo-controlled trial for which the power of the trial was determined by the primary endpoint (change from baseline in foot pain as measured by the NPRS). These analyses were intended to generate further hypotheses regarding the underlying causes, contributing factors and other details related to the clinical presentation in HV. The relationships between baseline HV or IM angle, and pain, as measured in the intention-to-treat population (all randomized participants) by NPRS, mFFI, and SF-36, were assessed using Pearson's correlation coefficient. Subgroup analyses used a mixed model for repeated measures on change from baseline in the daily scores averaged over the 7 consecutive days prior to a given study visit. Visit (weeks 4, 8, and 12), treatment-group-by-visit interaction, the stratification parameter (laterality [unilateral vs bilateral HV] at baseline), the subgroup, the subgroup by treatment and/or visit interaction, and the baseline value were included in the model. The proportion of participants experiencing clinically meaningful pain relief was assessed post hoc, defined as the proportion who reported an NPRS score in the 7 days prior to each study time point that was lower than their lowest baseline daily NPRS score.

Results

Participant Disposition and Baseline Characteristics

Full participant disposition was described previously (24). A total of 531 participants were screened and 186 were enrolled between June 2018 and May 2020. Mean (standard deviation [SD]) age was 48.2 (13.1) years and 91.9% (171/186) of participants were female (Table 1). Most participants (66.7%; 124/186) had bilateral HV and 17.7% (33/186) used orthotics. Mean (SD) time from diagnosis to first assessment was 6.4 (8.7) years. Mean (SD) HV and IM angles were 20.7° (5.2°) and 11.9° (2.4°), respectively. Mean (SD) baseline pain scores were 6.9 (1.6) using the NPRS, 63.8 (16.0) on the mFFI pain subscale, and 52.6 (23.1) using the SF-36 bodily pain domain. Mean (SD) mFFI baseline disability and activity scores were 56.8 (18.9) and 23.7 (23.1), respectively.

Pain and Angular Deviation

No relationship was observed between foot pain (n = 185 observations in each correlation) and HV angle (NPRS: r = 0.09, p = .23; mFFI: r = 0.09, p = .20; Fig. 2A, B), or IM angle (NPRS: r = 0.03, p = .64; mFFI: r = 0.04, p = .57). Furthermore, no relationship was observed between baseline SF-36 bodily pain domain score (n = 181 observations) and HV (r = -0.19, p < .05; Fig. 2C) or IM angle (r = -0.07, p = .36).

Disability/Activity Limitation and Angular Deviation

At baseline, no relationships (n = 185 observations in each correlation) were observed between HV or IM angle, and disability (HV angle r = 0.12, p = .12; IM angle r = 0.04, p = .59) or activity limitation scores (HV angle r = 0.14, p = .06; IM angle r = 0.05, p = .47). SF-36 subscales, including physical functioning, did not correlate with either HV or IM angle (data not shown).

Impact of Orthotic Use on Pain Relief

Participants who did (n = 33) and did not (n = 153) use orthotics during the trial reported similar mean (SD) NPRS scores at baseline (placebo: 6.3 [1.6] and 6.6 [1.4]; aboBoNT-A 300U: 6.7 [1.8] and 7.3 [1.6]; aboBoNT-A 500U: 6.5 [1.0] and 6.9 [1.7]; respectively). In the placebo and aboBoNT-A 300U groups, mean (standard error [SE]) changes from baseline in NPRS scores were also similar between participants who did and did not use orthotics, with the greatest changes observed at week 8 (placebo: -2.06 [0.69] and -2.03 [0.35]; aboBoNT-A 300U: -1.80 [0.70] and -1.70 [0.35]; respectively; Table 2). With aboBoNT-A 500U, the greatest changes from baseline were observed at week 8 for participants using orthotics (-2.50 [0.94]) and at week 12 for those who did not use orthotics (-2.53 [0.35]). With numbers available, no statistically

Table 1

Participant demographic and disease characteristics

Characteristic	All Participants (N = 186)	
Age, mean (SD), years	48.2 (13.1)	
Female, <i>n</i> (%)	171 (91.9)	
Time since diagnosis, mean (SD), years	6.4 (8.7)	
0 days, n (%)	29 (15.6)	
>0-2 years, n (%)	48 (25.8)	
>2-5 years, n (%)	28 (15.1)	
>5 years, n (%)	71 (38.2)	
Missing	10 (5.4)	
NPRS score, mean (SD)	6.9 (1.6)	
<7, n (%)	94 (50.5)	
≥7-9, <i>n</i> (%)	69 (37.1)	
>9, n (%)	22 (11.8)	
Missing	1 (0.5)	
mFFI subscales, mean (SD)		
Pain	63.8 (16.0)	
Disability	56.8 (18.9)	
Activity	23.7 (23.1)	
SF-36, mean (SD)		
Bodily pain	52.6 (23.1)	
General health	79.8 (14.4)	
Mental health	78.2 (17.1)	
Physical functioning	68.1 (24.8)	
Role emotional	83.3 (23.9)	
Role physical	67.9 (28.0)	
Social functioning	79.4 (23.8)	
Vitality	64.2 (18.8)	
Mental component	53.6 (9.3)	
Physical component	45.6 (8.7)	
HV angle, mean (SD), °	20.7 (5.2)	
≤20°, n (%)	89 (47.8)	
>20°, n (%)	97 (52.2)	
IM angle, mean (SD), °	11.9 (2.4)	
≤12°, n (%)	114(61.3)	
>12°, n (%)	72 (38.7)	
Orthotic use		
Yes	33 (17.7)	
No	153 (82.3)	
HV status, n (%)		
Unilateral	62 (33.3)	
Bilateral	124 (66.7)	

Data for the ITT population are presented.

Abbreviations: HV, hallux valgus; IM, intermetatarsal; ITT, intention-to-treat; mFFI, modified Foot Function Index; NPRS, Numeric Pain Rating Scale; SD, standard deviation; SF-36, 36-item Short-Form Health Survey.

significant differences were observed with aboBoNT-A 300U or 500U compared with placebo with or without orthotics at any time point; a trend toward improved efficacy was observed at week 12 with abo-BoNT-A 500U compared with placebo in participants who did not use orthotics (p = .0746; Table 2).

The impact of orthotic use was also evaluated using a different measure of treatment benefit in which clinically meaningful pain relief was assessed. The proportion of participants treated with aboBoNT-A 500U who achieved a meaningful response at week 12 was greater in the subgroup of participants who did not use orthoses (51.3% [20/39]) than those who did (16.7% [1/6]; Fig. 3A), with no difference between subgroups observed in the placebo group (21.1% [8/38] no orthoses vs 22.2% [2/9] orthotic use).

Impact of Laterality on NPRS Score and Pain Relief

Baseline NPRS scores (mean [SD]) were similar between participants with unilateral (n = 62) and bilateral (n = 124) HV (aboBoNT-A 500U: 6.57 [1.68] vs 6.98 [1.67]; aboBoNT-A 300U: 7.14 [1.54] vs 7.16 [1.70]; placebo: 6.80 [1.48] vs 6.43 [1.36]; respectively). LS mean (SE) change from baseline to week 12 in NPRS score was considerably greater in the

B. Baravarian et al. / The Journal of Foot & Ankle Surgery 00 (2023) 1-7



Fig. 2. Lack of relationship between HV angle and participant-reported pain measured with (*A*) NPRS, (*B*) mFFI pain subscale, and (*C*) SF-36 bodily pain domain. Pearson's correlation coefficient was used. (*A*) and (*B*) *N* = 185 observations. (*C*) *N* = 181 observations. HV, hallux valgus; mFFI, modified Foot Function Index; NPRS, Numeric Pain Rating Scale; SF-36, 36-item Short-Form Health Survey.

Table 2

Impact of orthotic use on change from baseline in NPRS score

Time Point	Orthotic Use			No Orthotic Use		
	Placebo ($n = 13$)	AboBoNT-A 300U (<i>n</i> = 13)	AboBoNT-A 500U (<i>n</i> = 7)	Placebo ($n = 50$)	AboBoNT-A 300U (<i>n</i> = 50)	AboBoNT-A 500U (<i>n</i> = 53)
Week 4						
LS mean change from baseline (SE)	-1.12 (0.63)	-1.06 (0.67)	-2.46(0.84)	-1.83 (0.33)	-1.43 (0.33)	-1.74(0.32)
p value vs placebo	-	.5266	.1008	-	.8086	.5785
Week 8						
LS mean change from baseline (SE)	-2.06(0.69)	-1.80(0.70)	-2.50(0.94)	-2.03 (0.35)	-1.70(0.35)	-2.38 (0.34)
p value vs placebo	-	.6041	.3501	-	.7448	.2369
Week 12						
LS mean change from baseline (SE)	-1.36 (0.70)	-1.42 (0.70)	-1.73 (0.94)	-1.81 (0.36)	-1.67 (0.35)	-2.53 (0.35)
p value vs placebo	-	.4756	.3752	-	.6087	.0746

Abbreviations: AboBoNT-A, abobotulinumtoxinA; LS, least-squares; NPRS, Numeric Pain Rating Scale; SE, standard error.

aboBoNT-A 500U and placebo groups for participants with unilateral HV than with bilateral HV (aboBoNT-A 500U: -3.06 [0.58] vs -1.99 [0.39]; placebo: -2.45 [0.53] vs -1.19 [0.39]; respectively). Results in the abo-BoNT-A 300U group were similar between participants with unilateral and bilateral HV (-1.74 [0.54] and -1.48 [0.38], respectively).

When the proportion of participants achieving a meaningful response was considered by laterality of HV, the placebo effect was similar in participants with unilateral HV compared with bilateral HV (26.3% [5/19] vs 17.9% [5/28], respectively; Fig. 3B). However, in the aboBoNT-A 500U arm, a considerably greater proportion of participants with unilateral HV achieved a meaningful response than participants with bilateral HV (73.3% [11/15] vs 33.3% [10/30], respectively; Fig. 3B).



The day-to-day variability of pain severity as reported by individual participants at baseline was assessed for impact on pain reduction 12 weeks after treatment. An SD of >2 for baseline NPRS scores was considered indicative of high pain variability, and a minimal clinically important difference (MCID) of 1.5 was calculated based on MCID values reported via NPRS in postbunionectomy trials (29,30). Variable pain reporting during the 7-day baseline period was observed in a total of 14 participants (placebo, n = 5; aboBoNT-A 300U, n = 7; aboBoNT-A 500U, n = 2), and, of these, 11 (placebo, n = 4; aboBoNT-A 300U, n = 6; aboBoNT-A 500U, n = 1) did not show improvements in NPRS score beyond an MCID at week 12 (Fig. 3C).



Fig. 3A. Proportion of participants achieving clinically meaningful pain relief at week 12, by orthotic use.^a



Fig. 3B. Proportion of participants reaching a clinically meaningful pain relief at week 12, by laterality.^a



Fig. 3C. Individual NPRS score reporting at week 12 for participants with at least 4 NPRS score values at both baseline and week 12.^b

^aData are presented for the percentage of participants in each subgroup with an NPRS score lower than the minimum NPRS score recorded in the 7 days prior to baseline for all 7 days prior to week 12.

^bBoxed data indicate variable reporters that are insensitive to change. Dashed line indicates a cut-off of 2 for standard deviation. Solid purple lines indicate the MCID interval.

AboBoNT-A, abobotulinumtoxinA; MCID, minimal clinically important difference; NPRS, Numeric Pain Rating Scale.

Impact of Angle Severity and Chronicity of HV on NPRS Score

No impact on treatment outcome was observed based on the severity of HV or IM angle at baseline. LS mean (SE) change from baseline NPRS score to week 12 in participants receiving aboBoNT-A 500U with an HV angle of $\leq 20^{\circ}$ was -2.33 (0.44) compared with -2.52 (0.48) in participants with an HV angle >20° (placebo: -1.92 [0.45] and -1.52 [0.45], respectively). LS mean (SE) change from baseline NPRS score to week 12 in participants receiving aboBoNT-A 500U with an IM angle of $\leq 12^{\circ}$ was -2.27 (0.42) compared with -2.62 (0.52) in participants with an IM angle of $>12^{\circ}$ (placebo: -1.80 [0.40] and -1.55 [0.54], respectively). No statistically significant differences between mean NPRS scores were observed with aboBoNT-A 500U compared with placebo at any time point regardless of HV or IM angle, although a trend toward efficacy was observed at week 12 in participants with an HV angle of >20° or an IM angle of >12° (p = .0653 and p = .0773, respectively).

Further to this, baseline pain severity (NPRS) did not impact treatment outcome. However, there was a numerical trend for greater improvement in participants with more severe baseline NPRS scores in the aboBoNT-A 500U group; this trend was mirrored in the placebo group. LS mean (SE) change from baseline to week 12 in participants receiving aboBoNT-A 500U was -1.67 (0.47), -2.71 (0.53), and -4.02(0.89) in participants with an NPRS score of <7, \geq 7-9, or >9 at baseline, respectively (placebo: -0.81 [0.41], -2.47 [0.55], and -4.67 [1.39], respectively). No statistical difference was observed with aboBoNT-A 500U compared with placebo at any time point, although a trend toward efficacy was observed with aboBoNT-A 500U compared with placebo at week 12 in participants with a baseline NPRS score of <7 (p = .0815).

Chronicity (i.e., the amount of time participants have lived with an HV diagnosis) was also found to have no impact on change in pain severity following treatment. LS mean (SE) change from baseline NPRS score to week 12 in participants receiving aboBoNT-A 500U was -2.34 (0.87), -2.16 (0.64), -2.47 (0.96), and -2.74 (0.49) in participants who received a diagnosis at 0 days (n = 8), 0 to 2 years (n = 15), 2 to 5 years (n = 8), and >5 years (n = 26) prior to study enrollment, respectively (placebo: -1.34 [0.68], -1.86 [0.66], -1.67 [0.76], and -2.29 (0.56); n = 14, 15, 11, and 20; respectively). With numbers available, no significant differences were observed with aboBoNT-A 500U compared with placebo at any time point.

Discussion

Emphasis on angular deviation in HV as a determinant of intervention and prognosis has increased in recent years among foot and ankle specialists, in part owing to payor considerations that often require radiographic evidence of minimum angular deviations to approve treatments (12-14). Basing these access decisions on angular deviation alone obviates the impact of the chronic underlying forefoot pain associated with HV even in patients with angular deviations which do not yet warrant surgical correction. There remains a general lack of consensus (based on available data) regarding the relationship between angular deviation in HV and pain or disability associated with the deformity. Other disease factors, such as the extent of bilateral disease, the day-today variability of pain severity, concomitant use of orthotic devices, severity of HV and IM angular deviation, pain severity, and chronicity, may impact the efficacy of treatments for pain. We conducted post hoc analyses of data from a large study that evaluated the efficacy and safety of aboBoNT-A compared with placebo as treatment for HV-associated pain in patients who had not yet undergone surgical correction, specifically to explore these factors. These presenting factors may be less clinically apparent than the centrally mediated pain state changes that occur following treatment, but may be important to consider when determining prognosis.

In this study, there was a near complete absence of a relationship between baseline HV or IM angle and participant-reported pain at baseline when measured using the NPRS, which was corroborated using 2 alternative pain measures. This suggests that a relationship between pain and angular deviation in HV is not a universal clinical reality, and that angular deviation alone should not be used to make prognostic and treatment decisions. To our knowledge, this is the first evidence from a large, randomized trial showing no correlation between HV or IM angular deviation and severity of pain using a numeric rating scale. These data add to evidence from previous studies that showed no relationship between HV or IM angle and foot pain as measured using a foot and ankle outcome instrument, SF-36, or a visual analog scale (2,14,31,32). Together, these data indicate that pain status should be an important clinical consideration complementary to the angle of deviation when making treatment decisions related to HV. Conversely, a recent large (N = 512) study of HV measuring pain using a visual analog scale reported a positive correlation between HV angle severity and pain, with the second and third metatarsal bones reported being the main regions of pain, regardless of HV severity (33).

In other studies, an association was reported between HV angle and pain when using the Manchester Foot Pain and Disability Index, a foot health status questionnaire and/or the SF-36 "Bodily Pain" subscale (10,34,35). A study in participants with HV found a strong correlation between physical ability and pain, as recorded with the Foot and Ankle Ability Measure and the Patient-Reported Outcome Measurement Information System (36). In the present study, in addition to assessing the relationship between pain and angular deviation, the relationship between the degree of disability and angular deviation, as well as the degree of activity limitation and angular deviation was evaluated. HV and IM angle showed no correlation with disability scores. Activity limitation scores also did not show an association with HV or IM angle in the present study. This was in line with previous studies using the Foot and Ankle Outcome Score or foot health functional status questionnaire (14,31), but in contrast with another study that utilized the Foot Function Index to measure activity limitation (10). We believe these numerous discordant studies highlight the complex interplay between structural deformity in HV, and shoe gear, pain, and perceived functional limitation, and we look forward to further efforts that explore this domain.

In these post hoc analyses, more participants in the aboBoNT-A 500U group who did not use orthoses reported clinically meaningful pain relief than those who did use orthotics, and than participants in the placebo and aboBoNT-A 300U groups regardless of orthotic use. It is feasible that participants who used orthoses had, historically, more severe pain than participants who did not use them, which was reduced

by orthotic use (because these subgroups had similar baseline NPRS scores) and dampened the subsequent response to aboBoNT-A treatment. More participants with unilateral HV reported improvements in pain than participants with a bilateral HV diagnosis, with almost a fullpoint difference in mean NPRS score change from baseline between groups. Participants with less variable pain scores at baseline were more likely to report marked pain relief following aboBoNT-A treatment than those with more variable baseline scores, suggesting that patients with high pain variability are not likely to benefit from treatment for pain, though participant numbers were small. In the present study, the majority of highly variable pain reporters were randomized to receive placebo or aboBoNT-A 300U, which may have resulted in more pronounced placebo and nonspecific pain relief responses in these treatment groups as reported in other studies (37,38), leading to a dampening of the pain relief effect observed with aboBoNT-A 500U. Chronicity of HV and the severity of angular deviation did not impact treatment outcome in this study.

Together, these data suggest that HV-associated pain and disability are key presenting factors that should be considered and clinically evaluated independently of angular deviation, and should not be inferred from radiographic results because they do not appear to correlate with HV or IM angle. Consequently, our results support evidence that a lack of correlation between pain and angle may have implications for HV treatment and reimbursement. Some insurance providers require a minimum angle of deviation for corrective HV surgery (12); and may need to focus less on HV/IM angle, and instead prioritize the severity of patient pain and disability as critical factors for either conservative or operative treatment of HV. Laterality, pain variability, and orthotic use should also be taken into account, and may be more important factors to consider than chronicity, or severity of HV/IM angle and pain in determining prognosis. However, further research is required to assess the impact of these factors on presentation of HV and treatment outcome to determine the clinical characteristics of patients who are most likely to experience pain relief with pharmacological treatments such as aboBoNT-A (5).

Although this study was not powered to formally analyze the impact of laterality, pain variability, and orthotic use on pain reporting and treatment outcomes, in these post hoc analyses, potentially clinically relevant differences in sensitivity to treatment effects were observed between subgroups, particularly following aboBoNT-A 500U treatment (for which the greatest pain relief effect was observed in this phase 2 study), and warrant further investigation. Over-the-counter orthotics were permissible only if used for the 30 days prior to screening, resulting in low participant numbers for this subgroup. Additionally, custom-made and dynamic orthotics have been shown to reduce pain in HV more effectively than over-the-counter or static products (39,40), and their inclusion in this study may have further pronounced the separation between the "with" and "without" orthotics groups in terms of treatment response. There are also a number of negative sequelae associated with HV in addition to pain that are important components of the patient experience that were not part of the present analyses, including impact on sleep and the psychological impact of visual deformity.

In conclusion, the results of these post hoc analyses of a large clinical trial in participants with HV who had not undergone surgery suggest that there is little rationale to exclusively consider HV or IM angles to make treatment decisions, particularly surgical correction, because the results of these analyses showed a lack of association between pain and angular deviation in patients with HV and mild to moderate severity forefoot pain. Pain status and its confounding factors should be considered equally relevant to the degree of angular deviation in clinical decision-making for the treatment of HV. This has substantial implications for the standard of care diagnosis and treatment of HV in terms of medical training, acceptable standards of the timing of medical intervention, and how payors consider coverage of HV-related pain and disability in

clinical practice and the impact of their sequelae, to address the individualized needs of patients with HV better.

Data Sharing

Anonymized patient-level study data that underlie the results reported in this publication may be made available to researchers who provide a research proposal. Additional relevant study documentation such as the clinical study report, study protocol with any amendments, statistical analysis plan may also be made available. Patient level data will be anonymized, and study documents will be redacted to protect the privacy of study participants. Where applicable, data from eligible studies are available 6 months after the studied medicine and indication have been approved in the USA and EU or after the primary manuscript describing the results has been accepted for publication, whichever is later. Further details on Ipsen's sharing criteria, eligible studies, and process for sharing are available at https://vivli.org/mem bers/ourmembers. Any requests should be submitted to www.vivli.org for assessment by an independent scientific review board.

Authors' Contributions

Substantial contributions to study design and data acquisition: B.B., L.A.D., M.V., D.G.A., and R.S. Data analysis or interpretation: M.V. Drafting of the publication or revising it critically for important intellectual content: all authors. Final approval of the publication: all authors.

Ethics Statement

The study was approved by appropriate health authorities and by independent ethics committees/institutional review boards, and was conducted in accordance with the Declaration of Helsinki, the International Conference on Harmonization Consolidated Guideline on Good Clinical Practice and all local regulatory guidelines. Written informed consent was provided by all participants prior to enrollment.

Acknowledgments

The authors thank Nicola Winstone, DPhil, and Malgorzata Urbacz of Ashfield MedComms, an Ashfield Health company, for providing medical writing and editorial support, which was sponsored by Ipsen, Slough, UK, in accordance with Good Publication Practice guidelines. The authors also thank all participants involved in the study, as well as their caregivers, care team, and the investigators and research staff who comprised the DYSTANCE study team.

References

- 1. Ayub A, Yale SH, Bibbo C. Common foot disorders. Clin Med Res 2005;3:116.
- Hurn SE, Vicenzino BT, Smith MD. Correlates of foot pain severity in adults with hallux valgus: a cross-sectional study. J Foot Ankle Res 2014;7:32.
- 3. Nix S, Smith M, Vicenzino B. Prevalence of hallux valgus in the general population: a systematic review and meta-analysis. J Foot Ankle Res 2010;3:21.
- Arinci Incel N, Genç H, Erdem HR, Yorgancioglu ZR. Muscle imbalance in hallux valgus: an electromyographic study. Am J Phys Med Rehabil 2003;82:345.
- Perera AM, Mason L, Stephens MM. The pathogenesis of hallux valgus. J Bone Joint Surg Am 2011;93:1650.
- Takakura Y, Tanaka Y. Pathogenesis of hallux valgus. In: Hirohata K, Mizuno K, Matsubara T, eds. *Trends in Research and Treatment of Joint Diseases*, Tokyo: Springer Japan, 1992; 149.
- 7. Wülker N, Mittag F. The treatment of hallux valgus. Dtsch Arztebl Int 2012;109:857.
- Klugarova J, Janura M, Svoboda Z, Sos Z, Stergiou N, Klugar M. Hallux valgus surgery affects kinematic parameters during gait. Clin Biomech (Bristol, Avon) 2016;40:20.
- 9. Clarke GR, Thomas MJ, Rathod-Mistry T, Marshall M, Menz HB, Peat G, Roddy E. Hallux valgus severity, great toe pain, and plantar pressures during gait: a cross-sectional study of community-dwelling adults. Musculoskeletal Care 2020;18:383.
- Coşkun G, Talu B, Bek N, Bayramlar KY. Effects of hallux valgus deformity on rear foot position, pain, function, and quality of life of women. J Phys Ther Sci 2016;28:781.

B. Baravarian et al. / The Journal of Foot & Ankle Surgery 00 (2023) 1-7

- Roddy E, Zhang W, Doherty M. Prevalence and associations of hallux valgus in a primary care population. Arthritis Rheum 2008;59:857.
- AETNA. Simple Bunionectomy Policy. 2019. Available at: http://www.aetna.com/cpb/ medical/data/600_699/0629.html. Accessed January 14, 2022.
- UnitedHealthcare Oxford. Policy: Bunionectomy (or hallux valgus) Surgery. 2005. Available at: https://www.oxhp.com/providers/toolsResources/practicalResources/ policy/bunionectomy.html. Accessed January 14, 2022.
- Matthews M, Klein E, Acciani A, Sorensen M, Weil L, Weil LS, Fleischer A. Correlation of preoperative radiographic severity with disability and symptom severity in hallux valgus. Foot Ankle Int 2019;40:923.
- Ferrari J, Higgins JP, Prior TD. Interventions for treating hallux valgus (abductovalgus) and bunions. Cochrane Database Syst Rev 2004:1. Cd000964.
- Torkki M, Malmivaara A, Seitsalo S, Hoikka V, Laippala P, Paavolainen P. Surgery vs orthosis vs watchful waiting for hallux valgus: a randomized controlled trial. Jama 2001;285:2474.
- 17. Ferrari J. Bunions. BMJ Clin Evid 2009;2009:1112.
- Radović PA, Shah E. Nonsurgical treatment for hallux abducto valgus with botulinum toxin A. J Am Podiatr Med Assoc 2008;98:61.
- Wu KP, Chen CK, Lin SC, Pei YC, Lin RH, Tsai WC, Fuk-Tan Tang S. Botulinum toxin type A injections for patients with painful hallux valgus: a double-blind, randomized controlled study. Clin Neurol Neurosurg 2015;129(suppl 1):S58.
- Moghtaderi A, Dehghan F, Mousavizadeh A, Khakpour N. Evaluation of the therapeutic effect of botulinum toxin a on hallux valgus deformity and pain. Int J Phys Med Rehabil 2017;5(6).
- Mittal SO, Jabbari B. Botulinum neurotoxins and cancer a review of the literature. Toxins (Basel) 2020;12:32.
- Oh HM, Chung ME. Botulinum toxin for neuropathic pain: a review of the literature. Toxins (Basel) 2015;7(6).
- Winner BM, Bodt SML, McNutt PM. Special delivery: potential mechanisms of botulinum neurotoxin uptake and trafficking within motor nerve terminals. Int J Mol Sci 2020;21:8715.
- 24. Armstrong DG, DiDomenico LA, Baravarian B, Parekh SG, Volteau M, Silva R. Pain reduction with abobotulinumtoxina for the treatment of hallux valgus in adult participants: results of a randomized and placebo-controlled phase 2 trial. J Foot Ankle Surg 2022. https://doi.org/10.1053/j.jfas.2022.07.002. Jul 12:S1067-2516(22)00211-3.
- 25. Parekh SG, Armstrong DG, DiDomenico LA, Baravarian B, Volteau M, Silva R. AbobotulinumtoxinA in the management of hallux valgus in adult patients: results of a randomized and placebo-controlled phase II trial. In: Poster Presented at 1st Virtual EFORT Congress, Oct 28-30, Virtual Meeting; 2020.
- Contopoulos-Ioannidis DG, Karvouni A, Kouri I, Ioannidis JP. Reporting and interpretation of SF-36 outcomes in randomised trials: systematic review. BMJ 2009;338:a3006.
- Dawson J, Doll H, Coffey J, Jenkinson C. Responsiveness and minimally important change for the Manchester-Oxford foot questionnaire (MOXFQ) compared with

AOFAS and SF-36 assessments following surgery for hallux valgus. Osteoarthritis Cartilage 2007;15:918.

- Laucis NC, Hays RD, Bhattacharyya T. Scoring the SF-36 in orthopaedics: a brief guide. J Bone Joint Surg Am 2015;97:1628.
- 29. Riff DS, Duckor S, Gottlieb I, Diamond E, Soulier S, Raymond G, Boesing SE. Diclofenac potassium liquid-filled soft gelatin capsules in the management of patients with postbunionectomy pain: a Phase III, multicenter, randomized, double-blind, placebo-controlled study conducted over 5 days. Clin Ther 2009;31:2072.
- Daniels SE, Baum DR, Clark F, Golf MH, McDonnell ME, Boesing SE. Diclofenac potassium liquid-filled soft gelatin capsules for the treatment of postbunionectomy pain. Curr Med Res Opin 2010;26:2375.
- Badlissi F, Dunn JE, Link CL, Keysor JJ, McKinlay JB, Felson DT. Foot musculoskeletal disorders, pain, and foot-related functional limitation in older persons. J Am Geriatr Soc 2005;53:1029.
- Thordarson DB, Ebramzadeh E, Rudicel SA, Baxter A. Age-adjusted baseline data for women with hallux valgus undergoing corrective surgery. J Bone Joint Surg Am 2005;87:66.
- Chang C, Wang QF, Guo JC, Li DD, Fan YB, Wen JM. The biomechanical relationship between hallux valgus deformity and metatarsal pain. J Healthc Eng 2020;2020:8929153.
- 34. Cho NH, Kim S, Kwon DJ, Kim HA. The prevalence of hallux valgus and its association with foot pain and function in a rural Korean community. J Bone Joint Surg Br 2009;91:494.
- Menz HB, Roddy E, Thomas E, Croft PR. Impact of hallux valgus severity on general and foot-specific health-related quality of life. Arthritis Care Res (Hoboken) 2011;63:396.
- 36. Nixon DC, McCormick JJ, Johnson JE, Klein SE. PROMIS pain interference and physical function scores correlate with the Foot and Ankle Ability Measure (FAAM) in patients with hallux valgus. Clin Orthop Relat Res 2017;475:2775.
- 37. Farrar JT, Troxel AB, Haynes K, Gilron I, Kerns RD, Katz NP, Rappaport BA, Rowbotham MC, Tierney AM, Turk DC, Dworkin RH. Effect of variability in the 7-day baseline pain diary on the assay sensitivity of neuropathic pain randomized clinical trials: an ACT-TION study. Pain 2014;155:1622.
- Harris RE, Williams DA, McLean SA, Sen A, Hufford M, Gendreau RM, Gracely RH, Clauw DJ. Characterization and consequences of pain variability in individuals with fibromyalgia. Arthritis Rheum 2005;52:3670.
- 39. Chadchavalpanichaya N, Prakotmongkol V, Polhan N, Rayothee P, Seng-Iad S. Effectiveness of the custom-mold room temperature vulcanizing silicone toe separator on hallux valgus: a prospective, randomized single-blinded controlled trial. Prosthet Orthot Int 2018;42:163.
- Arias-Martín I, Reina-Bueno M, Munuera-Martínez PV. Effectiveness of custom-made foot orthoses for treating forefoot pain: a systematic review. Int Orthop 2018;42:1865.